Author’s response to reviews

Title: Trend of disparity between coastland and inland in medical expenditure burden for rural inpatients with malignant tumor in southeast of China from 2007 to 2016

Authors:

Rong Fu (fjctfr@sina.com)
Zheng Lin (pglin@msn.com)
Fei He (x117x@163.com)
Yixian Jiang (371466977@qq.com)
Zhenquan Zheng (zzquan229@163.com)
Zhijian Hu (huzhijian@fjmu.edu.cn)

Version: 5 Date: 22 Feb 2020

Author’s response to reviews:

Editor Comments:

1. Authorship Change Form

We note that the order of authors has changed since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

In such cases, we use a standardised form which we would request you and your co-authors to complete. The authorship change form can be found at the following link: https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5

All instructions can be found on the form, please treat the 'current authorship' section as the original authorship. Please return the form within 14 days by email to the Editorial Office with all author signatures (including those newly added/removed).

Response 1: We will return the form by email to the Editorial Office with all author signatures as soon as possible.

2. Consent
Please clarify if the patients gave their oral consent to participate. Currently we note “patients were given oral informed consent that their medical record data may be used for academic research”

Response 2: We have revised the manuscript.

3. Clean manuscript

As a reminder, at this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response 3: We have kept the rules.